Reviewer’s code: 06129102

SPECIFIC COMMENTS TO AUTHORS
In this retrospective study, the authors discussed the effectiveness of coronary injection with nitroprusside, a commonly used vasodilator, as a pharmacological intervention strategy administered before balloon inflation for the prevention of no reflow or slow flow during percutaneous coronary intervention. The results showed preventive use of sodium nitroprusside proved beneficial. The methods are clearly described, and the results are very interesting. The reviewer suggests to accept it after a minor revision. 1. Some minor language polishing should be corrected. 2. The title is too long, please short it. 3. A conclusion should be added. 4. The references should be edited according to the journal’s guideline.

Answer: Thank you very much for your comment.
1. Some minor language polishing should be corrected.
The language was proofed carefully.

2. The title is too long, please short it.
The title was shorted.

3. A conclusion should be added.
A conclusion was added.

4. The references should be edited according to the journal’s guideline.
The references were updated according to the guideline.